Author's response to reviews

Title: Rationale, design and methodology of a double-blind, randomized, placebo-controlled study of escitalopram in prevention of Depression in Acute Coronary Syndrome (DECARD)

Authors:

Baiba Hedegaard Hansen (baiba.hansen@gmail.com)
Jamal Abed Hanash (jamal.a.hanash@gmail.com)
Alice Rasmussen (alice.ras@dahlnet.dk)
Jørgen Fischer Hansen (jfh01@bbh.regionh.dk)
Morten Birket-Smith (mbs01@bbh.regionh.dk)

Version: 4 Date: 30 March 2009

Author's response to reviews:

Page 12

Secondary (safety) endpoints

15-minute day and night HRV